Author's response to reviews

Title: A case of a four-year-old child adopted at eight months with unusual mood patterns and significant polypharmacy.

Authors:

Magdalena Romanowicz (romanowicz.magdalena@mayo.edu)
Jennifer Vande Voort (VandeVoort.Jennifer@mayo.edu)
Alastrair McKean (mckean.alastair@mayo.edu)

Version: 3 Date: 31 Aug 2017

Author’s response to reviews:

Thank you so much for your comments and suggestions. We have addressed all of them, please see answers highlighted below.

a. Title page:

Please include a title page at the front of your manuscript file. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

Title page included

b. Declarations section

Please ensure that the sections,

- Ethics approval and consent to participate (please state 'Not applicable' in the section)
- Funding
- Acknowledgements
- Authors' Information (optional)

are included at the end of your manuscript in your Declarations section. Please include all sections, and indicate in the text those that do not apply by stating 'Non Applicable'.

That has been added

c. Table 1:
We note that the Table 1 has been included in duplicate. Please remove the duplicate Table from your revised manuscript.

We removed the duplicate table

d. Reviewer reports:

Matthew Krasowski (Reviewer 1): The authors have adequately addressed my critiques. The revisions in response to critiques from other reviewers are also acceptable.

Steven Pliszka (Reviewer 2): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.

Revisions are satisfactory, no further edits requested.

Glad to hear that all our revisions were satisfactory. Thank you.